Author's response to reviews

Title: Mycoplasma hominis brain abscess after a head trauma: A case report

Authors:

Andrés F Henao-Martínez (andres.henaomartinez@ucdenver.edu)
Heather Young (andres.henaomartinez@ucdenver.edu)
JJ Loes Nardi-Korver (andres.henaomartinez@ucdenver.edu)
William Burman (andres.henaomartinez@ucdenver.edu)

Version: 2 Date: 21 March 2012

Author's response to reviews:

Changes were made to conform the formatting style of the journal.